

This is a repository copy of *Happier than them, but more of them are happy: Aggregating subjective well-being.*

White Rose Research Online URL for this paper: http://eprints.whiterose.ac.uk/145081/

Version: Published Version

Monograph:

Sechel, C. orcid.org/0000-0002-8653-3443 (2019) Happier than them, but more of them are happy: Aggregating subjective well-being. Working Paper. Sheffield Economic Research Paper Series, 2019008 (2019008). Department of Economics , University of Sheffield. ISSN 1749-8368

© 2019 The Author(s). For reuse permissions, please contact the Author(s).

Reuse

Items deposited in White Rose Research Online are protected by copyright, with all rights reserved unless indicated otherwise. They may be downloaded and/or printed for private study, or other acts as permitted by national copyright laws. The publisher or other rights holders may allow further reproduction and re-use of the full text version. This is indicated by the licence information on the White Rose Research Online record for the item.

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.





The University Of Sheffield. Department Of Economics.

Sheffield Economic Research Paper Series.

Happier Than Them, but More of Them Are Happy: Aggregating Subjective Well-Being

Authors Cristina Sechel

ISSN 1749-8368

SERPS no. 2019008

April 2019

www.sheffield.ac.uk/economics

Happier than Them, but More of Them Are Happy: Aggregating Subjective Well-Being

Cristina Sechel University of Sheffield, UK <u>c.sechel@sheffield.ac.uk</u>

April 2019

Abstract

This paper proposes the use of headcount-based indicators for the measurement of national Subjective Well-Being (SWB). It provides a methodological contribution to the challenge of threshold selection for headcount measures using Cognitive Dissonance Theory operationalised using life satisfaction data from World/Eurpean Values Surveys. A Beta-regression approach is employed to explore the empirical relationships between national SWB and objective measures of well-being contributing to the empirical literature on social indicators. The use of this model is novel in this context. The findings reveal relationships between objective measures of development and SWB that are not apparent when average national SWB is used. For example, I find no significant link between national income and the share of satisfied individuals.

Keywords. Subjective Well-Being; Cognitive Dissonance Theory; Beta-regression. **JEL classifications:** O1, I3, H1.

Acknowledgements

A special thank you goes to Prof. Karen Mumford for helpful discussions and comments. I would also like to thank Prof. Mozaffar Qizilbash for suggestions during the early stages of this paper, and Prof. Sarah Brown for her valuable feedback during the Royal Economic Society Women's Committee Mentoring Meetings. This version has benefited from comments of participants in the 18th IZA European Summer School in Labor Economics, and the 2014 Wellbeing and Public Policy Conference. All errors are my own.

1 Introduction

Subjective measures of well-being have recently motivated a great deal of research within economics. Mounting evidence suggests that subjective well-being (SWB) data are reliable and valid sources of information (Diener, 1994; Kesebir and Diener, 2008) that can effectively supplement standard objective indicators (Frey and Stutzer, 2014; Graham, 2008). Several studies highlight the benefits of constructing and maintaining national accounts of SWB for use in conjunction with objective measures (Bruni et al., 2008; Cummins, 2016; Diener and Seligman, 2004; Diener and Suh, 1997; Fleurbaey, 2009; Stiglitz et al., 2010), while some go as far as to advocate the use of SWB as the one single measure of progress (Layard, 2009). Furthermore, there are several attempts to build fundamental guidelines for potential measures of national SWB (Cummins et al., 2003; Diener, 2006).

The operationalization and construction of aggregate measures of SWB is currently a subject of considerable interest in sociology and statistics (e.g. Casacci and Pareto, 2018; Maggino, 2009), but there has been little effort within economics to develop normative frameworks and analysis methods for aggregate measures of SWB. The economic literature to date has generally relied on traditional methods (which have previously been used for conventional income-based aggregate indicators of development) and is largely limited to one single measure of national life satisfaction, namely the mean. However, the simple mean is not a suitable metric of aggregate satisfaction considering the characteristics of SWB data, which are captured by scales that are arbitrary, ordinal and bounded. Bond and Lang (2014) have recently criticized the mean-based approach to comparing satisfaction (or happiness)¹ across countries, and more generally across groups of people.

Given the nature of subjective data, the construction of aggregate measures of subjective wellbeing requires careful balance between accuracy and the level of detail. Headcount measures may prove to be more suitable indicators of overall well-being since they are less informationally demanding. While average measures impose strong assumptions on the underlying SWB scale (e.g. cardinality), headcount measures require only an ordinal comparison below and above a certain cut-off point. Although this class of measures does not utilize the full range of information contained in SWB scales, this paper puts forth the argument that the resulting aggregate information it provides is more reliable relative to mean measures. Specifically, it is potentially preferable to rely on an indicator based on binary information than one based on a higher resolution scale when that scale is potentially inaccurate and misleading.

¹ The terms 'satisfaction' and 'happiness' are generally considered to measure different aspects of SWB, though they are sometimes used interchangeably (Diener, 2006; Easterlin, 2004). However, the terms 'happiness' and 'happy' are often used to refer to SWB broadly (as in Happiness Economics) and do not imply any specific aspect of SWB. This paper focuses specifically on reported life satisfaction as opposed to reported levels of happiness (see Section 3), but does use the terms 'happiness' and 'happy' to refer to SWB in general.

The principal challenge with headcount measures is identifying a relevant and meaningful cutoff value. The foremost aim of this paper is to develop a methodological approach for constructing a headcount measure of the proportion of individuals who report above a specific threshold value of SWB. The approach applies Cognitive Dissonance Theory (Akerlof and Dickens, 1982) to survey data from the World Values Survey and the European Values Survey to identify a meaningful threshold above which individuals are considered to be (sufficiently) satisfied with life. This provides both a theoretical framework and real-world relevance, and contributes to the economics literature on cross-country analysis of well-being and to measurement theory. It can also be applied outside of economics where the focus has been primarily on multidimensional indices of well-being (see for example Fattore et al., 2015; Mauro et al., 2016), not on threshold selection. Compared to more complex statistics, the proposed headcount has the practical advantage that it is easy to understand and relate to for policymakers and the public.

The second contribution of this paper is to identify an appropriate empirical model for estimating the relationships between the proposed headcount measure and standard objective indicators of development. I extend the econometric analysis in previous studies of national SWB, which generally relies on Ordinary Least Squares (OLS) (such as Deaton, 2008; Ovaska and Takashima, 2006; Stevenson and Wolfers, 2008), by using a Beta-regression model which takes into account the skewed and naturally bounded nature of the distribution of reported SWB data. Beta-regression is shown to be more suitable given the distinct properties of SWB data. The emphasis on standard objective indicators is deliberately chosen because of the strong influence they exert on how we view development. The concern is that these conventional accounts help create a shared view that may be easily misguided if the measures it relies on do not adequately reflect overall well-being.

The paper is intended as a starting point for discussion about best methods of aggregating subjective information and theory-based approaches to headcount measures of SWB. It shows that different national measures of SWB can convey diverse stories about development and well-being. Choosing the appropriate aggregation method is therefore crucial for effective policy design.

The remainder of the paper is structured as follows: Section 2 discusses the relevant literature; Section 3 describes the methodology for constructing the proposed alternative aggregate of SWB; Section 4 presents an empirical application using data from the World Values Survey and the European Values Survey; and Section 5 concludes.

2 Literature

Initial studies at the national level focused on simple correlations between income and SWB (Easterlin, 1974). These were soon followed by a growing body of literature encompassing various objective accounts of well-being, including development measures beyond income-based indicators, such as life

expectancy, educational attainment, health indicators, female labour participation, economic and political freedoms, to name a few (Blanchflower and Oswald, 2005; Deaton, 2008; Lawless and Lucas, 2011; Leigh and Wolfers, 2006; Ovaska and Takashima, 2006).

The SWB literature relies heavily on simple average measures, lacking consideration for alternative non-utilitarian approaches to national SWB. Non-mean based aggregation procedures, such as the headcount measure of the share of satisfied individuals proposed in this paper, have been used for simple descriptions of datasets (Oswald, 1997) but not as key measures of interest in international accounts of development. While Easterlin (1974) does take into account some distributional considerations², his main cross-country result is based on average happiness, as are subsequent studies concerned with national SWB, including Easterlin's more recent work on the happiness paradox (Easterlin et al., 2011) and Stevenson and Wolfers's treatment of Easterlin's findings (Stevenson and Wolfers, 2008).

Notable exceptions are the 'happy life expectancy' measure proposed by Veenhoven (1996) and a measure of satisfaction with life that is not explained by personal characteristics (Di Tella et al., 2001). The former is defined as the product of standard life expectancy and average happiness (transformed on a scale ranging from 0 to 1); the latter is the average of the residuals obtained by regressing individual-level life satisfaction on personal characteristics. These measures show more sophisticated alternatives for aggregating self-reported well-being, but they nevertheless rely on average SWB and are utilitarian³ in nature.

A direct reference regarding a headcount measure of national SWB can be found in Helliwell and Huang (2008), who briefly mention using "the share of respondents above or below particular cutoff points in the numerical distribution of responses" (p. 609). The aim of their paper is to assess the effect of the quality of government on national life satisfaction. The share is used as a robustness check for differences in the shape of the distribution of satisfaction responses due to cultural differences. This differs in intent from the current study, which aims to explicitly develop a methodology for creating a headcount measure as a national indicator of aggregate SWB. Helliwell and Huang (2008) find no significant changes in the key findings when using the share measure, but the relevant results are not reported in the publication, and no specific cut-offs are discussed.

² Summary statistics of the distribution of SWB are considered, but only for happiness questions with qualitative scales involving limited categories (e.g. 'very happy', 'fairly happy', 'not very happy')

³ In general, utilitarian approaches refer to utility maximization. Mean measures are linked specifically to average utilitarianism, where average utility is maximized.

3 A Headcount aggregate of national subjective well-being

Mean measures of individual-level SWB information require precise interpersonal comparisons, but the ordinal and arbitrary nature of reported SWB scales makes it difficult to compare answers across individuals. Bond and Lang (2014) show that cross-country comparisons of average SWB are virtually impossible when reported SWB scales are ordinal (without imposing strong assumptions about the underlying distributions of SWB). Furthermore, SWB scales are naturally bounded, which limits the growth of average SWB measures since individuals who have reached the highest level cannot improve further.

Perhaps more critical than data structure and interpretation are normative considerations. Mean measures promote average utilitarianism, and this may be a misguided social aim. Frankfurt (1987) reasons that what is most important morally in terms of the distribution of economic assets is "that each [person] should have *enough*" (p. 21, italics in original). This principle is even more salient when measuring SWB because its complex nature makes it unreasonable to expect perpetual increases in average levels. Given that SWB depends on many life dimensions – some of which governments cannot or should not have control over – it is perhaps more appropriate for governing bodies to target some reasonable standard of SWB for all citizens, rather than seek to continually increase the well-being of all.

A sufficientarian welfarist approach provides a fitting alternative to the utilitarian-based mean measures of SWB, and is well suited for use with SWB information. Unlike average utilitarianism, which seeks to maximize average welfare, sufficientarianism is primarily concerned with providing a 'sufficient' level of welfare. Crisp (2003) proposes that "compassion for any being B is appropriate up to the point at which B has a level of welfare such that B can live a life which is sufficiently good" (p. 762). In terms of subjective welfare, development can accordingly be viewed as a nation's ability to support such a sufficient level of SWB for its citizens, or (crucially) for as many of its citizens as possible. The obvious question arises regarding the *sufficient* level of SWB. This is the main challenge of this paper and will be addressed later in this section after introducing the general form of the proposed headcount ratio and the underlying SWB information.

Applying the sufficiency principle to SWB data translates to an aggregate measure that is based on a dichotomous reduction of self-reported well-being and can be expressed formally as follows:

$$SWB_{share} = \frac{1}{n} \sum_{j=1}^{n} I(swb_j \ge z)$$
⁽¹⁾

Where *n* is the total number of individuals in a country, swb_j is individual *j*'s reported level of SWB, *z* is a threshold level of welfare and I(.) is an indicator function that is 1 when individual *j*'s reported SWB is above the threshold level *z* and 0 otherwise. The threshold level, *z*, separates individuals who

have a reasonably high level of SWB from those who do not. SWB_{share} therefore represents the proportion of individuals in the country with a sufficiently high level of SWB.

The scale and scope of swb_j can vary depending on the data source. There are several types of questions currently used in various surveys, broadly classified into two groups: happiness and satisfaction⁴. Happiness questions are considered measures of emotional states, whereas satisfaction measures are viewed as cognitive evaluations (Ovaska and Takashima, 2006). The former can be problematic because they tend to elicit more hedonic evaluations that reflect mainly current (recent) mood. On the other hand, life satisfaction questions are more appropriate for national measures of wellbeing – Helliwell and Barrington-Leigh (2010)⁵ argue that they are "more reflective of overall and continuing life circumstances and hence are more suited to capture long-term and international differences in policies and institutions" (p. 732). This is key for the construction of SWB_{share} , which is intended as an overarching measure that can capture broad evaluations about life in general across countries. I therefore construct SWB_{share} based on reported life satisfaction data, such that:

$swb_i = s_i$

where s_j is the reported life satisfaction of individual *j*. *SWB*_{share} is suitable for use with bounded and ordinal scales, and more importantly, has limited sensitivity to small differences in reported life satisfaction so it addresses to some degree the problem of interpersonal comparisons by reducing the number of comparisons made between adjacent points on the satisfaction scale⁶. We are only concerned with comparing satisfaction levels around *z* so that all reported satisfaction levels below the threshold are assumed to denote lower SWB than all satisfaction levels above the threshold. As such, the value of *SWB*_{share} as a measure of overall national SWB hinges on the choice of *z*, which separates individuals who are sufficiently satisfied from those who are not.

The range of SWB in another important consideration for constructing the share of satisfied individuals. National and international surveys typically use satisfaction scales ranging from 4 to 11 points⁷. It is generally accepted that questions with finer scales are more reliable (Diener et al., 2009; Helliwell and Barrington-Leigh, 2010). The proposed headcount measure should therefore be based on SWB questions with a high resolution scale. As such, subsequent analysis in this paper is based on a 10-point scale (see Section 4.1).

⁴ See Bruni and Porta (2007) for further discussion on the different aspects of SWB.

⁵ There are also questions regarding specific aspects of life (e.g. satisfaction with the freedom to choose how to live one's life, satisfaction with the educational system, satisfaction with the quality of air, etc.) but these do not adequately reflect life in general.

⁶ This level of interpersonal comparison is supported by Diener and Tov (2012): "For example, a person reporting an '8' on a 10-point happiness scale is virtually always happier than someone who reports a '3', and the latter is much more likely to suffer from clinical depression. However, a person who reports a '7' might not invariably be less happy than a person who reports an '8"" (p. 12).

⁷ For a more detailed summary of the various SWB questions and scales used in a variety of surveys see Diener (1994).

I now return to discuss the challenge of choosing the level of sufficiency. The answer lies in finding a point which holds special meaning for most individuals, one which we are psychologically inclined to regard as an important and meaningful threshold. In order to find such a point, it is important to remember that reported satisfaction values reflect the interplay between one's life and one's feelings about that life. A theory that recognizes this underlying relationship is necessary to understand the satisfaction profile of individuals.

Cognitive Dissonance Theory, first proposed by Hirschman (1965), provides an intuitive framework for understanding this relationship. According to Hirschman, dissonance occurs when our view of ourselves does not match the reality of our actions. It is uncomfortable and undesirable, and we aim to minimize the level of dissonance in our lives. The central significance of Hirschman's theory is that it suggests that dissonance is more easily reduced by changing views rather than changing actions. Sometimes, it is only possible to alter views if the actions in question have already taken place.

Akerlof and Dickens (1982) later proposed that dissonance often occurs because our view of ourselves as "smart, nice people" is challenged by the reality of past actions or new information. In the context of SWB, I propose that we like to think of ourselves as being satisfied/happy with our lives, at least on some basic level. There are two opposing forces at work: (i) a strong resistance against admitting a less than some acceptable level of satisfaction because we seek to uphold this view of ourselves as satisfied; while (ii) cognitive dissonance pushes us to admit our true level of satisfaction.

To clarify, let us consider the implications along the SWB path. A positive relationship between life conditions and SWB means that poorer life conditions lead to lower life satisfaction since we seek to minimize dissonance between reported satisfaction and life conditions. When life conditions are acceptable, we have no problem correctly identifying the appropriate satisfaction level. However, this relationship breaks down temporarily around a threshold that we consider to represent a 'basic' level of happiness because there is a reluctance to admit satisfaction levels below this point. In terms of the distribution of reported life satisfaction, we should see a pile-up of responses at this threshold. This is where dissonance builds up as the disparity between life conditions and SWB increases, eventually forcing individuals to adjust their view of themselves as happy/satisfied and thus report levels of satisfaction below this key threshold. Dissonance therefore peaks around this resistance threshold as shown in Figure 1.

[Figure 1 about here]

This approach requires the use of survey data in order to examine the distribution of satisfaction levels. The next section demonstrates how this can be done in practice and applies the proposed headcount measure of SWB to analyse the relationships between national SWB and standard objective indicators of development. As outlined in Section 2, studies typically assess the cross-country associations of mean life satisfaction with Gross Domestic Product (Easterlin et al., 2011) and various

social indicators (Blanchflower and Oswald, 2005; Deaton, 2008; Lawless and Lucas, 2011; Leigh and Wolfers, 2006; Ovaska and Takashima, 2006). The next section extends this literature by exploring the association between the proposed headcount measure, SWB_{share} , and several development measures.

4 Application: life satisfaction and objective well-being across countries

4.1 Data

Self-reported life satisfaction data from World Values Surveys (WVS, 2009) and European Values Surveys (EVS, 2011) are used to construct SWB_{share} . These initiatives are conducted independently but are compatible and comparable across countries and time, and are available in an integrated dataset that is well-documented. In general, WVS and EVS cover different countries but there is some overlap. There are a total of 6 WVS and 4 EVS waves (each conducted over multiple years)⁸ based on repeated cross-sections of stratified random samples of around 1,500 individuals per country on average. Early waves are limited in their coverage and do not include representative samples for some countries. Since this is a cross-country examination, the analysis is restricted to waves 3 and 4 of WVS and waves 3 and 4 of EVS to maximize country coverage. To construct a national-level panel dataset with comparable country sets and similar time frames, waves 3 of WVS and EVS are combined into one 6-year period covering 1999-2004, and waves 4 are combined into another 6-year period covering 2005-2010⁹. Over these two periods, there are a total of 253,010 respondents from 99 countries; of which 14,768 are excluded due to incomplete information (either pertaining to life satisfaction or to the country-level explanatory variables introduced below). The analysis is based on a sample of 141 observations (136 for one of the specifications) for a total of 90 countries, with 78 country-observations in period one and 63 in period two¹⁰.

Respondents are asked "*All things considered, how satisfied are you with your life as a whole these days?*" and are instructed to choose a number between 1 and 10, where 1 is labeled "dissatisfied" and 10 is labeled "satisfied"¹¹. WVS and EVS data are suitable for implementing a practical proxy for the share of satisfied individuals: the high-resolution scale is more reliable compared to coarser alternatives, and they support rich cross-country analyses of SWB since they cover a large and

⁸ WVS waves were conducted in 1981-84, 1989-1993, 1994-1999, 1999-2004, 2005-2007 and 2010-2014. EVS waves were conducted in 1981-1984, 1989-1993, 1999-2004 and 2008-2010.

⁹ See Table A1 in the Appendix for a list of countries and their availability in each wave.

¹⁰ Previous country-level studies on SWB have used samples ranging from 44 to 166 observations (Deaton, 2008; Di Tella et al., 2001; Leigh and Wolfers, 2006; Ovaska and Takashima, 2006; Stevenson and Wolfers, 2008).

¹¹ Except for wave 2005-2007 of the WVS in which 1 means "completely dissatisfied" and 10 means "completely satisfied".

representative set of countries ranging from underdeveloped to fully industrialised economies, representing all continents and major sub-regions. The distribution of life satisfaction responses for the sample used in the regression analysis is shown in Table 1 for each period separately (counts have been adjusted using sampling weights provided with the survey data).

[Table 1 about here]

I use these reported life satisfaction data to identify a meaningful satisfaction threshold z and construct SWB_{shar} . The distribution of responses is characterized by a pronounced data-cliff between levels 4 and 5 (Figure 2). Satisfaction levels of 5 or higher are consistently more prevalent than levels below 5 in each of the two periods. This pattern is also observed for most of the countries individually (graphs not shown but available upon request). Moreover, satisfaction level 5 is more prevalent than the adjacent level 6. Taken together with the sharp data-cliff described above, this suggests a pile-up effect at 5 and a marked reluctance to report below this point. This pattern matches the cognitive dissonance argument presented in Section 3, indicating level 5 may be a meaningful threshold for interpreting satisfaction responses as the point where dissonance is highest. In this framework, individuals reporting below level 5 can be interpreted to have such poor life conditions that they cannot overcome the instinct to deny that they are indeed not within the acceptable range of satisfaction. As such, level 5 may be set as z, the lowest point at which people are sufficiently satisfied, and the proposed alternative headcount measure of national SWB from equation 1 can be more precisely defined as:

$$SWB_{share} = \frac{1}{n} \sum_{j=1}^{n} \theta_j I(s_j \ge 5)$$
⁽²⁾

where s_j is individual *j*'s life satisfaction response (in this case) ranging from 1 to 10, and θ_j is respondent *j*'s sample weight included in order to obtain results representative of the whole population.

[Figure 2 about here]

The analysis is based on regressing SWB_{share} on the individual components making up the current formulation of the Human Development Index (HDI)¹²: per capita Gross National Income (GNI), life expectancy, mean years of schooling, and expected years of schooling. The choice of these components is motivated by the HDI's widespread popularity within the current discourse on development and social progress. Some previous studies of national SWB relied on variations of these measures to help explain

¹² For more details on the HDI components see Technical Notes of the Human Development Report (UNDP, 2016) available at <u>http://hdr.undp.org/sites/default/files/hdr2016_technical_notes.pdf</u>.

mean reported life satisfaction (Deaton, 2008; Easterlin et al., 2011; Leigh and Wolfers, 2006; Stevenson and Wolfers, 2008).

Data for the HDI components are taken from the online database maintained by the United Nations Development Programme (UNDP, 2013)¹³. Matching these indicators to satisfaction data by specific year is not possible since the Values Surveys are conducted in waves that span multiple years. Additionally, yearly UNDP data are not available prior to 2005. For countries surveyed in the period 1999-2004, I use UNDP data for year 2000. For countries surveyed in the period 2005-2010, I use averages of UNDP data over those years. Period-averages are considered to produce results that reflect a more long-term relationship with subjective measures (McGillivray, 2005); this method has been used previously in order to minimize seasonal deviations from the long-term trend (Ovaska and Takashima, 2006). The current study is concerned mainly with international comparisons and therefore with fundamental differences in the economic organization of the countries, which are slow to change, so measures capturing a long-term trend are ideal. All development indicators and additional control variables are defined in Table 2.

[Table 2 about here]

Per capita GNI is expected to have a strong positive effect on the share of satisfied individuals. Economists often rely on the assumption that income and well-being are directly linked: "there is a clear presumption that changes in economic welfare indicate changes in social welfare in the same direction, if not in the same degree" where national product "is taken to be the objective, measurable counterpart of economic welfare" (Abramovitz, 1959, p. 3). Previous studies find a positive relationship between national income and SWB measured using mean life satisfaction (Diener and Oishi, 2000; Easterlin et al., 2011; Ouweneel and Veenhoven, 1991; Stevenson and Wolfers, 2008). The effect of life expectancy on SWB is ambiguous. Living longer can reflect a society in good health, which boosts life satisfaction, but this effect can be reduced or reversed if the quality of life for the old is low. Previous evidence is contradictory – Ovaska and Takashima (2006) find a positive relationship between life expectancy and life satisfaction, while Deaton (2008) estimates a negative link. We should also see a positive association between national satisfaction and mean/expected years of schooling. According to the human capital model, education is positively linked to increased welfare through a positive effect on wages. However, the empirical evidence is mixed. A number of studies find a positive association between education and reported life satisfaction (Blanchflower and Oswald, 2004; Easterlin, 2001; Ferrer-i-Carbonell, 2005; Graham and Pettinato, 2002), while others find a negative or statistically insignificant relationship (Blanchflower and Oswald, 2005; Flouri, 2004; Powdthavee, 2008; Shields et

¹³ Available online at <u>http://hdr.undp.org/en/statistics/data/</u> (accessed on Sept. 4, 2012). UNDP does not directly collect data; their database is constructed using various sources (list of sources available at <u>http://hdr.undp.org/en/statistics/understanding/sources/</u>).

al., 2009). Powdthavee et al. (2015) argue that there are a number of non-financial channels through which education effects well-being. These channels can mitigate the effect of education on well-being independently of income gains. They find a negative direct association between years of education and life satisfaction, but positive indirect associations through income, employment, marital status and health. This suggests that the overall relationship between education and life satisfaction may be positive or negative depending on which channels have a stronger impact on personal well-being. Bjornskov et al. (2008) propose that the relationship between education and SWB is stronger in low-income countries where the income gains of extra schooling are larger. This suggests that there may be considerable variation in the way populations react to gains in knowledge across the countries analysed in this paper.

Cultural norms and social systems vary widely across nations and they can be systematically and significantly related to individuals' assessment of their own life satisfaction. A concern is that many cultural dimensions tend to be highly correlated with standard objective measures of well-being, especially with income (e.g. individualistic, democratic countries also tend to be the richest and most developed). I control for cultural differences using the Inglehart and Welzel (2010) two-dimensional index. The index is constructed using responses to attitudinal questions from WVS and EVS. Nations are scored along a traditional vs. secular-rational value scale, and also along a survival vs. selfexpression value scale. Both scales revolve around zero so that cultures that emphasize traditional and survival values are assigned negative scores, while those with emphasis on secular-rational and selfexpression values are given positive scores. Figure 3 shows the position of each country in the sample along these two cultural dimensions. Cultural profiles vary greatly across the nations in the sample, spreading across much of the bi-dimensional value plane.

[Figure 3 about here]

Cultural value scores are averages between available country scores from 1999-2004 and 2005-2010 (i.e. if scores are available for both waves, then the average is used, otherwise a single score value is used)¹⁴. This ensures that all countries in the sample are assigned one score (for each dimension) that does not change over time¹⁵.

There are several advantages to using the Inglehart-Welzel indices to control for cultural effects¹⁶. Firstly, they are directly relevant to the SWB data used here given they are themselves based

¹⁴ Except for Armenia, Azerbaijan, Georgia, and Uganda, for which no score data are available between 1999-2010. Earlier information prior to 1999 is used for these countries.

¹⁵ The decision to average across both time-periods for those countries for which both data points are available was made because few countries are given scores in both time periods and also to reduce bias stemming from large differences in cultural profiles for countries that significantly change their values and attitudes between wave 1 and 2.

¹⁶ Alternative cultural controls have been used in the literature. For example, Deaton (2008) uses a binary indicator variables to identify eastern European and sub-Saharan countries. Ovaska and Takashima (2006) single-out Asian

on information collected by the WVS and the EVS. Secondly, they are systematically constructed using Factor Analysis of responses to questions explicitly designed to capture cross-national differences in value-systems and to gain a better understanding of cultural distinctions. Lastly, the two dimensions provide simple, reduced-form controls that capture wide-ranging aspects of values and beliefs.¹⁷

I also consider an extended model that includes unemployment and inflation data from the World Development Indicators database (WDI, 2014) to control for macroeconomic forces. Both national unemployment and inflation rates are known to have a strong negative affect on life satisfaction (Blanchflower et al., 2014; Di Tella et al., 2003). Moreover, unemployment has been shown to mitigate the benefits of attained education on life satisfaction (Powdthavee et al., 2015) so a high unemployment rate is likely to have an impact on the relationship between mean years of schooling and the share of satisfied individuals.

Summary statistics for all the variables used in the analysis are presented in Table 3. The share of satisfied individuals ranges from 39% to 98% and it is on average higher in 2005-2010 (85%) than in 1999-2004 (79%). All of the objective well-being indicators are also higher in period two on average: per capita GNI rises from 15,100 to 17,548, life expectancy increases from 71.89 to 73.81, mean years of schooling increases from 8.38 to 9.03, and expected years of schooling increases from 13.20 to 13.82. Lastly, the average unemployment and inflation rates are 8.86 and 6.87. It is worth noting that there are large country differences across all of these indicators.

[Table 3 about here]

4.2 Econometric Model

The baseline econometric model commonly used in the literature to explore the relationship between objective and subjective indicators of well-being can be expressed as:

$$SWB_i = \alpha + \beta' X_i + \varepsilon_i, \quad i = 1, \dots, N$$
(3)

countries and include religion controls for Islam and Christianity. Helliwell (2003) classifies countries into six groups: industrial, former Soviet Union, other Eastern European countries, Latin American, Asian, other developing countries and Scandinavian. These approaches do not have any underlying value systems and thus have no intuitive interpretation other than to identify differences between groups of countries.

¹⁷ Detailed information regarding the variables used to construct the two dimensions and their correlations is available online as a supplementary material to Inglehart and Welzel (2010) at http://journals.cambridge.org/pps2010020.

where SWB_i is typically average life satisfaction for country *i*, but can also be an alternative measure such as mean happiness or annual change in life satisfaction (Easterlin, 2013), and *X* is a vector of objective well-being measures¹⁸. In this paper, SWB_i is SWB_{shar} for country *i*.

This can be estimated using Ordinary Least Squares; however, OLS regression can produce fitted values that are outside the bounds of SWB_{share} , and does not address the non-normal distribution of this SWB_{share} . Figure 4 shows that the proportion of satisfied individuals is left-skewed in each of the two constructed waves, with most countries concentrated at the upper end of the distribution and with long left tails. Under the OLS normality assumption this asymmetry can lead to misleading inference about the statistical characteristic of the estimates¹⁹.

[Figure 4 about here]

Ferrari and Cribari-Neto (2004) and Smithson and Verkuilen (2006) independently propose a Betaregression model for skewed, naturally bounded dependent variables. They specifically develop these models for use with bounded scales from survey responses and proportions. The Beta function allows great flexibility in modelling asymmetric distributions, and Beta models perform well with small datasets (Kieschnick and McCullough, 2003). The following Beta-regression model with a Logit link function is estimated using the data described in Section 4.1:

$$E(SWB_{it}|X_{it}) = \frac{e^{\beta' X_{it}}}{1 + e^{\beta' X_{it}}}$$
(4)

where SWB_{it} is SWB_{share} in country *i* at time period *t*, and *X* includes the following explanatory and control variables: *PER CAPITA GNI*, *LIFE EXPECTANCY*, *MEAN YEARS OF SCHOOLING*, *EXPECTED YEARS OF SCHOOLING*, a binary time indicator that equals 1 for observations in the second wave and 0 for observations in the first wave. Additional extensions also include the two Inglehart and Welzel cultural indices, unemployment and inflation as controls. *PER CAPITA GNI* is measured in constant 2005 international dollars and is logarithmically transformed²⁰. $E(SWB_{it}|X_{it})$ is the conditional mean of SWB_{share} , and β is a matrix of parameter vectors to be estimated. Following the notation in Ferrari and Cribari-Neto (2004), SWB_{share} is assumed to be distributed with a beta density given by:

¹⁸ This simple model has been largely applied to cross-sectional data (Deaton, 2008; Leigh and Wolfers, 2006); in some cases a cross-section is constructed by averaging across a number of waves to minimize seasonal deviations from the long-term trend (Ovaska and Takashima, 2006). Stevenson and Wolfers (2008) use a wide range of data sources and waves to analyze both cross-section and panel datasets.

¹⁹ More precisely, the normality assumption refers to the conditional distribution of the dependent variable (i.e. conditional on the regressors). Smithson and Verkuilen (2006) point out that linear regression is often robust to violations of this assumption, but they emphasize that this is not always the case, especially in small samples, and is particularly misleading for "survey responses with bounded response sets or proportions" (p. 54).

²⁰ The relationship between income and SWB is better captured by a logarithmic scale (Helliwell, 2003).

$$f(SWB_{share};\mu,\phi) = \frac{\Gamma(\phi)}{\Gamma(\mu\phi)\Gamma((1-\mu)\phi)} (SWB_{share})^{\mu\phi-1} (1-SWB_{share})^{(1-\mu)\phi-1}$$
(5)

where $\mu = E(SWB_{shar})$ is the mean, ϕ is a precision parameter, $\Gamma(\cdot)$ is the gamma function, and $0 < SWB_{share} < 1$. The model is estimated using Maximum Likelihood with panel-robust standard errors to control for heteroskedasticity and serial correlation within countries.²¹

The explanatory variables are collectively taken as proxies of human development, they are each intended to capture different objective aspects of development. However, there has been some criticism regarding the high internal correlation between these measures (Kovacevic, 2010), which can result in multicollinearity and lead to inflated standard errors with potentially biased estimates. It is not clear that multicollinearity is a serious concern in this case. Firstly, Kovacevic (2010) discusses this issue in detail and presents several sources that defend the choice of HDI components despite high correlations observed among them. Secondly, the Variance Inflation Factors (VIFs) are well below 10²². *PER CAPITA GNI* has the highest VIF value at 5.66, while the VIF values for *LIFE EXPECTANCY, MEAN YEARS OF SCHOOLING*, and *EXPECTED YEARS OF SCHOOLING* are 3.44, 2.78, and 4.44 respectively. These values indicate that correlations between the explanatory variables do not cause serious problems of multicollinearity.

The model does not establish causation, although there is some evidence that SWB is affected by education (Oreopoulos, 2003), and Frey and Stutzer (2002a) argue that it is affected by income, not vice-versa. However, Powdthavee (2010) finds evidence of endogeneity between income and life satisfaction. As in previous cross-country studies of SWB, the estimated marginal effects are interpreted as partial correlations between the covariates and the dependent variable. Nevertheless, establishing the presence of associations between SWB and objective measures of development is important for crosscountry comparisons as it helps us evaluate whether countries that are considered highly developed economically also exhibit high levels of SWB.

²¹ Given the panel structure of the data, a Fixed-Effects (FE) model was considered but not used due to the small sample. While minimizing bias, FE can be inefficient in small sample unbalanced panels. Furthermore, it is difficult to obtain consistent FE estimates in non-linear specifications such as the Beta-regresssion model proposed here (Cameron and Trivedi, 2009, p. 232). Consistency is also problematic in short panels (Cameron and Trivedi, 2009, p. 231). This problem is amplified here due to the panel being unbalanced with a considerable portion of countries appearing only in one of the waves. Of the total 90 countries included in the analysis, 12 only appear in the 1999-2004 wave and 27 only appear in the 2005-2010 wave, which leaves only 51 countries with enough information to compute the average values necessary for the FE estimators. Lastly, FE models are not a good choice when within-unit variance is much smaller than between-unit variance (Cameron and Trivedi, 2009), which is the case here (see Table A2 in Appendix). The Inglehart and Welzel (2010) cultural controls are used instead to control for country differences. This allows more degrees of freedom compared to the FE model. Helliwell (2003) use a similar approach in order to preserve degrees of freedom.

²² VIF values below 10 are considered to denote a degree of collinearity that is acceptable (e.g. Hair et al., 2006).

4.3 Results

Table 4 reports the marginal effects evaluated at the sample means of the regressors²³ estimated using Equation 4^{24} . The model presented in column 1 is the most parsimonious including only the individual HDI components and a wave dummy; cultural controls are added in column 2; and unemployment and inflation are added in column 3. The Bayesian Information Criterion (BIC)²⁵ is significantly lower than in the equivalent OLS specifications (shown in Appendix Table A3), which suggests that Beta-regression is superior at explaining the variation in the proportion of satisfied individuals than the standard OLS, as expected given the skewed distribution of *SWB*_{share}.

[Table 4 about here]

The parsimonious model in column 1 shows significant associations between the share of satisfied individuals and all objective measures of development. The marginal effects of *PER CAPITA GNI, MEAN YEARS OF SCHOOLING*, and *EXPECTED YEARS OF SCHOOLING* are all significant at the 1% level, while *LIFE EXPECTANCY* is significant at the 10% level. *PER CAPITA GNI* is positively linked with the share of satisfied individuals for the average country, but this relationship is small in magnitude. The marginal effect of 0.0568 indicates that a ten percent increase in *PER CAPITA GNI* from 10,543²⁶ is associated with an increase of approximately 0.00568 percentage points (pps) in the share of satisfied individuals. A more accurate estimate can be obtained by computing the share of satisfied individuals at GNI=\$11,597, which is 0.8422 and 0.8475 respectively (a 0.0053 pps change). The marginal effect is smaller at the mean of *PER CAPITA GNI* – a 10% increase from \$16,454 is associated with an increase of 0.0047 pps in the share of satisfied individuals. These differences are just under 4% of the standard deviation of the share of satisfied individuals.

²³ For a given continuous covariate x, the marginal effect at means is the partial derivative of the share of satisfied individuals with respect to x given by $\partial E(SWB_{share}|X)/\partial x = \partial \left(\frac{e^{X\beta}}{1+e^{X\beta}}\right)/\partial x$, evaluated at mean values of all the covariates in X. For a dichotomous covariate x, the marginal effect is the discrete change in the share of satisfied individuals as x changes from 0 to 1 given by $\frac{dE(SWB_{share}|X)}{dx} = E(SWB_{share}|X, x = 1) - E(SWB_{share}|X, x = 0).$

²⁴ Average marginal effects are generally larger in magnitude, but the results remain qualitatively the same (results available upon request).

²⁵ BIC is used because it allows for comparison across models with different dependent variables and different structural specifications. A lower value signals a better model fit. Following Kass and Raftery (1995), differences in BIC values that are less than 2 points constitute "very little" evidence to support the use of the model with the lower BIC value, while differences between 2 and 6 points constitute "some positive" evidence, differences between 6 and 10 constitute "strong" evidence, and differences larger than 10 present "very strong" evidence.

²⁶ This corresponds to the mean of $\ln(PER \ CAPITA \ GNI)$. Note that this is different from mean *PER CAPITA GNI* (which is 16,454). The mean of $\ln(PER \ CAPITA \ GNI)$ is used because the results reported in Table 3 are the marginal effects at the means of the covariates.

The marginal effects of the remaining objective indicators are as follows: a one year increase in *LIFE EXPECTANCY* from its mean value of 73 years is associated with an increase of 0.0029 pps in the share of satisfied individuals; a one year increase in *MEAN YEARS OF SCHOOLING* from its mean value of 8.7 years is associated with a decrease of 0.022 pps; and a one year increase in *EXPECTED YEARS OF SCHOOLING* from its mean value of 13.5 is associated with an increase of 0.013 pps.

Including the Inglehart-Welzel indices as cultural controls (column 2) improves the model fit (the BIC value is reduced by 63 points), and changes the significance of the marginal effects, though only the index of survival/self-expression values is statistically significant (at 1%). Countries that emphasize self-expression over survival values have a higher share of satisfied individuals. Most notably, PER CAPITA GNI is no longer significant at standard levels. The relationship between income and mean satisfaction is by comparison commonly found to be positive and statistically significant in cross-country analysis (e.g. Stevenson and Wolfers, 2008), which is a crucial part of the well-known Easterlin paradox²⁷. Consistent with previous studies, regressing mean life satisfaction using the same sample and models presented in Table 4 results in a positive marginal effect of PER CAPITA GNI that is statistically significant across all three specifications²⁸ (results shown in Appendix Table A4). The prominent role of income-based measures of development, both within and without economic studies, makes GNI a particularly important key measure of well-being. The non-significant marginal effect in column 2 may offer a novel perspective. The income-satisfaction relationship can be judged to be very different when national satisfaction is constructed to directly reflect the perceptions of the unsatisfied. This finding suggests evidence against the existence of trickle-down benefits - if trickle-down effects are strong then we might expect to see the same strong relationship between income and the share of satisfied individuals as we observe between income and mean satisfaction, but we do not, implying weak trickle-down effects on SWB. The cultural controls also decrease the marginal effects of MEAN YEARS OF SCHOOLING, and EXPECTED YEARS OF SCHOOLING, making the latter nonsignificant at standard levels, while increasing the marginal effect of *LIFE EXPECTANCY*.

The negative relationship between *MEAN YEARS OF SCHOOLING* and the proportion of satisfied individuals raises questions about the role of education within a SWB framework. It implies that adopting an account of progress based on SWB may lead to policies that do not support investing in education. This can be particularly detrimental for efforts to integrate SWB into accounts of well-being because it suggests an unpopular development agenda that would discourage education. However,

²⁷ Although the initial Easterlin paper found a weak and ambiguous relationship between income and happiness across countries (Easterlin, 1974), subsequent analysis shows a positive relationship and specifies that the paradox is characterized by the conflicting findings of (i) a positive income-happiness relationship *across and within* countries at a point in time, and (ii) a negative relationship across time within a given country (Easterlin et al., 2011).

²⁸ Beta-regression requires the dependent variable to be continuous and constrained on (0, 1). While the share of satisfied individuals naturally falls in this interval, mean satisfaction does not and is instead defined on (1, 10). Mean life satisfaction is therefore transformed such that $SWB'_{mean} = (SWB_{mean} - a)/(b - a)$, where a and b are the theoretical boundaries on SWB_{mean} (not the minimum and maximum observed in the sample).

this conclusion may be misguided, as there are likely to be many channels through which education can effect well-being. As discussed previously, Powdthavee et al. (2015) show that, despite a negative direct relationship, education and life satisfaction are positively associated indirectly through higher income, higher likelihood of employment, and better health. The negative marginal effect found here reflects the combined direct and indirect association between education and SWB, which potentially hides these benefits. For example, macro-economic conditions can reduce or even reverse any existing positive effects of education on satisfaction. Including unemployment rate and inflation (column 3)²⁹ gives a marginal effect of MEAN YEARS OF SCHOOLING that is non-significant at standard levels, while both macro controls are found to be statistically significant and negatively associated with the level of satisfaction. This result is consistent with the theory of adaptive expectations (Burchardt, 2005). A population that expects to achieve a high level of education is more likely to have increased expectations if people believe that better education will bring better opportunities and higher disposable income. If opportunities are subsequently not available, individuals are likely to feel let down after investing in education. This line of reasoning can explain the negative relationship between MEAN YEARS OF SCHOOLING and the proportion of satisfied individuals, which disappears when including the unemployment rate. These findings suggest that the potential benefits to education are closely linked to the availability of adequate post-education opportunities. This hypothesis resonates particularly well with the current economic conditions – large numbers of the educated youth (in both developed and developing nations) are underemployed and unhappy with their available employment prospects (House of Lords Committee EU, 2014).

Including unemployment and inflation has no substantial impact on the marginal effect of *PER CAPITA GNI* but does decrease the marginal effect of *LIFE EXPECTANCY* and increases that of *EXPECTED YEARS OF SCHOOLING*, making the latter statistically significant at the 10% level. There is no change in the sign or statistical significance of either of the cultural controls. It is worth noting that the significance of *EXPECTED YEARS OF SCHOOLING* is not contradictory to the non-significance of *MEAN YEARS OF SCHOOLING*. The latter measures the current stock of education, while the former measures the education of future generations. It is possible that current generations in countries with a higher level of education do not enjoy the direct well-being benefits due to, say, bad labour market conditions, but derive indirect benefits from a progressive education profile.

An advantage of the non-linearity of the Beta-regression model is that it can be used to compare marginal effects across different levels of the regressors which can offer valuable insights. For example, we can exploit this feature to test the presence of an income satiation point. Some scholars have suggested that there is a threshold level beyond which income does not improve well-being. This threshold may be relatively low, representing the amount of money required to secure a 'decent'

²⁹ This model includes 136 observations due to missing unemployment and/or inflation data for Andorra, Chile, Mali, Rwanda and Venezuela.

standard of living. Frey and Stutzer (2002b) find evidence that a threshold level exists at \$10,000, while Layard (2003) places it at \$15,000, though he more recently proposes \$20,000 (Layard, 2011). We can investigate how the relationship between *PER CAPITA GNI* and *SWB*_{share} differs across income levels by looking at the marginal effect path. Although the marginal effect is non-significant at the mean of *PER CAPITA GNI* in the preferred model (2), it may be the case that it is significant at lower levels of income, which would support the satiation point theory. Figure 5 shows that the marginal effect of ln(*PER CAPITA GNI*) diminishes only slightly as we move from countries with the lowest incomes to countries with the highest incomes. However, it is not statistically significant across all levels of income, so there is no evidence of a satiation point.

[Figure 5 about here]

It is interesting to note that the wave dummy is strongly significant and large in magnitude (relative to the other covariates) across all models in Table 4³⁰. It is associated with a 0.0313 pps increase in the share of satisfied individuals in the preferred specification (column 2). This indicates that reported SWB is improving over time, which presents a somewhat optimistic outlook for the future of social progress. More individuals seem to be happy with their lives more over time. While this does not help explain the process of improvement, it does suggest that we are moving toward a world-state that is more valuable to individuals. A positive interpretation is that these results reflect improved life circumstances in a progressive world. A more pessimistic view is that they could instead reflect lower expectations. A longer time-horizon and additional measures as more data become available will help answer this ambiguity.

I further explore this time dimension with differential effects across waves, by interacting wave with all covariates in the preferred model from column 2 in Table 4. Although the estimated marginal effects discussed above already vary by wave due to the non-linearity of the Beta model, the interaction effect allows for further variation that may not be captured in the baseline model. The corresponding marginal effects at means for each wave are shown in the first two columns of Table 5 (significant differences between waves are indicated in bold and italic). This reveals there is a positive link between income and SWB_{share} that is statistically significant in wave 2 but not in wave 1 (and this difference

³⁰ It is possible that the marginal effect of the wave dummy is biased due to the unbalanced structure of that data. If countries appearing only in the second wave are on average happier than countries appearing only in the first wave (all other regressors being held constant), the marginal effect will be biased upward. The share of satisfied individuals is on average higher for countries appearing only in the second wave, but so is *PER CAPITA GNI, LIFE EXPECTANCY* and both education measures. Excluding the time indicator does not significantly change the point-estimates of the key measures of interest. As a further robustness check, model 2 was repeated only for the subsample of countries that appear in both waves. The wave coefficient remains very strongly significant and similar in magnitude, and the results generally support those obtained using the full sample (the only difference is that expected school years becomes non-significant). The unbalanced structure of the panel does not appear to drive the strong positive time trend. A more detailed discussion regarding unbalanced panel issues follows below.

between the waves is statistically significant). Inversely, the marginal effect of *LIFE EXPECTANCY* is significant only in wave 1, as is the marginal effect of *MEAN YEARS OF SCHOOLING* (though the latter is not significantly different from wave 2). While these findings may be due to unobserved individual or country characteristics changing over time, it is possible that they are driven by the unbalanced sample, which I discuss in more detail below.

[Table 5 about here]

While the general purpose of cross-country analysis is to understand the overall relationships between objective measures of well-being on SWB, it may also be helpful for policy design to note whether there are any differences between these two groups. For example, the evidence so far suggests that the relationship between average income and average SWB is substantially diminished or non-existent for developed countries (see discussion in Section 3.3 of Frey and Stutzer, 2002b). To explore this further I include in the preferred specification (Table 4, column 2) a binary indicator for developed/developing nations (as recognized by the United Nations) interacted with all covariates and find contradictory evidence. The marginal effect of income on SWB_{share} is significantly larger for developed nations (and this difference is statistically significant), while the marginal effect of income for developing nations is not significantly different from zero at standard levels. However, the opposite is true for *MEAN YEARS OF SCHOOLING*, which has a significant marginal effect only for developing nations. It is also worth noting that the marginal effect of *LIFE EXPECTANCY* is only significant for developed nations, while the marginal effect of *EXPECTED YEARS OF SCHOOLING* is only significant for developed nations, though these differences are not statistically significant between the two groups of countries.

Lastly, I consider gender differences. Women report higher life satisfaction on average than men despite being worse off in terms of income and other life dimensions, though the reverse is true in some developing nations (Helliwell et al., 2012). In the sample used here there is no clear gender pattern across countries, about 45% of country-wave observations have a higher share of satisfied women relative to the share of satisfied men and this is stable across developed as well as developing nations. Little is known about how the relationship between subjective and objective measures of aggregate well-being differs between genders, which may have important policy implications. I explore this by running the preferred model separately for men and women where the dependent variable is the share of satisfied men and the share of satisfied women respectively, $SWB_{share}^g = \frac{1}{n}\sum_{j=1}^n \theta_j I(s_j \ge 5 | g)$ where g=(female, male). The results are shown in columns 5 and 6 of Table 5. I find a positive relationship between income and the share of satisfied women but no significant relationship for men. This suggests that women may be the primary beneficiaries of economic growth, which is consistent with diminishing marginal returns to income given that women have lower incomes than men on

average. Another notable difference between the genders is that *EXPECTED YEARS OF SCHOOLING* is positively related to aggregate SWB for men but not for women. It is not clear why this may be the case, but one explanation may be that men are more optimistic about future prospects (e.g. the returns to their children's education) than women.

Unbalanced Panel Issues

Unbalanced panels are common and can provide accurate estimates if the missing information is randomly distributed across the sample of relevant units. Common sources of unbalanced panels are attrition in respondents for surveys that follow the same individuals over a period of time, and shifting samples in rotating panel surveys. In this case, the missing information is not due to attrition (as macro-level panels do not rely on the retention of the same individuals, attrition is not generally applicable), and there is no clear intention from the part of the WVS and EVS for a systematic rotating panel design.

To investigate the validity of this type of unbalanced data, I begin by exploring the characteristics of the 39 countries that appear only in one of the two waves (referred to as single-wave countries) and how they behave relative to the rest of the sample. In general, the 12 countries appearing only in the first wave (call these group A) have on average lower values of *PER CAPITA GNI*, *LIFE EXPECTANCY*, and education measures compared to the first wave observations of countries that appear in both waves (at 5% statistical significance). The same is observed for the 27 countries that appear only in the second wave (call these group B) when compared to the second wave observations of countries that appear in both waves. However, these differences are not necessarily problematic in this case because the countries are both lost and added to the sample. As long as each separate wave contains a representative sample of countries, the random addition or loss of a group of countries should not bias the regression results. In other words, if group A is not significantly different from group B, the unbalanced structure of the dataset should not invalidate the results.

Moreover, t-tests confirm that all measures of interest are on average not significantly different between group A and B (at standard confidence levels), except for *EXPECTED YEARS OF SCHOOLING* (which is significant at the 10% level). This indicates that the addition and loss of countries across waves does not appear to drastically change the sample properties (i.e. seemingly similar countries are lost and gained). However, countries in the two subsamples may still exhibit very different relationships between regressors and the satisfaction measures, which is enough to introduce bias in the estimates. Comparing the results of the full sample with those of the restricted subsample of countries that appear in both waves is not particularly useful in this context because countries in groups A and B taken together may be different from countries that are surveyed in both waves, which would affect the estimation results without necessarily causing bias. The question is whether the addition of B is more or less equivalent to the loss of A. One way to test for this is to run separate regressions for each of the groups A and B to compare the resulting coefficients, but the small sample sizes make it difficult to obtain consistent estimates. Additional future waves will help settle this issue. However, it is possible to gain some insights by comparing the wave-specific marginal effects from the full sample (Table 5, columns 1 and 2) with the corresponding marginal effects from the restricted sample including only countries that are observed in both waves (Appendix Table A5). The results are qualitatively more consistent across waves in the balanced subsample with the exception of *EXPECTED YEARS OF SCHOOLING*, which is positively associated with SWB in wave 2 but not statistically significant in wave 1. In light of this, the full sample wave-specific results should be interpreted with caution as they may be biased due to the panel being unbalanced. In particular, there is no significant relationship between income and SWB_{sha} in either wave, which suggests that the positive association identified in wave 2 in the full sample (Table 5) is driven by countries that do not appear in both waves.

Data Comparability within Second Wave

There may be some concern about the general data comparability within the 2005-2010 period as some countries were surveyed prior to the 2008 recession, while others were surveyed after. If SWB is affected by the recession, aggregate measures of SWB in countries surveyed before 2008 may not be comparable with measures for countries surveyed after.

It is possible to explore the implications of this split sample using a subset of 20 countries surveyed by both initiatives in wave 2 using simple two-sample t-tests for the difference in the level of aggregate satisfaction between samples collected in 2005-2007 (by WVS) and those collected in 2008-2010 (by EVS). The results in Table 6 reveal that the share of satisfied individuals is significantly different between the EVS and WVS samples for 15 of the 20 nations, and mean satisfaction is significantly different for 13 nations, with both positive and negative differences. However, it is difficult to interpret these results as indicative of a recession effect because the changes observed by the t-tests may be caused by corresponding changes in other factors that are unaccounted for.

[Table 6 about here]

To gain further insight, a Chow test is performed on the baseline OLS model (for simplicity) to see how the estimates compare between the subsample of countries with WVS data and those surveyed only after the recession by EVS. The test reveals that the subsamples are significantly different at the 5% level, which is consistent with the above t-test results.

This issue can be further addressed by repeating the regressions using only the subset of 15 countries that are surveyed three times, once during the 1999-2004, once in 2005-2007, and again in 2008-2010. The use of the three periods allows for the estimation of a time trend before the recession, which helps to give relative meaning to the changes in satisfaction observed after the onset of the

recession. This country subset consists of a balanced panel with 45 country-period observations. Though this is a small subsample, it can help to get an impression of the impact of the recession. Regressing the share of satisfied individuals using model 2 from Table 4 and replacing the wave dummy with time indicators for 1999-2004 and 2008-2010 (the pre-recession period of 2005-2007 is omitted) shows an overall positive time trend. The marginal effect of the 1999-2004 period is -0.036 relative to 2005-2007 (significant at the 1% level), which means individuals are on average happier in 2005-2007 than in previous years. However, the positive marginal effects of 0.0013 for period 2008-2010 relative to 2005-2007 is not significant at standard levels. These findings indicate the presence of a negative recession effect on the life satisfaction of individuals which has substantially reduced the previously positive time trend. In light of this, the positive estimates presented above can be expected to be downwardly biased, while the negative estimates may be upwardly biased. For example, the positive estimate on *LIFE EXPECTANCY* may be significantly lower than it would otherwise have been in the absence of the recession.

5 Concluding Remarks

This paper aims to contribute to the well-being debate that is vital to the study of economics by developing a framework for measuring aggregate SWB. The approach employed applies Hirschman's Cognitive Dissonance Theory to widely available data on reported life satisfaction to construct a national headcount measure of national SWB. The intent has been to expand the tools available for SWB measurement in the context of economic growth and social progress, and to contribute to normative considerations regarding the social definition of SWB improvement.

The proposed headcount measure is based on a particular data-cliff between values 4 and 5 observed in the life satisfaction responses from the World and European Values Surveys. This cut-off point is interpreted to reflect the share of sufficiently satisfied individuals. However, the distribution of reported life satisfaction reveals that there are other notable points on the satisfaction scale that may be of interest for future research. In particular, there is a prominent peak at satisfaction level 8. While this additional information can potentially enrich our understanding of SWB, it is important to emphasize that the application of Cognitive Dissonance Theory to the sufficiency principle requires a level of satisfaction that is credibly low. It would be difficult to argue that level 8 (on a 1-10 satisfaction scale) should be interpreted as the level below which individuals may be considered to be *sufficiently* dissatisfied.

More generally, one concern regarding the use of threshold measures of SWB is their reliance on arbitrary cut-off values. Since subjective scales are not based on a set, measurable standard, choosing appropriate cut-off values is challenging. This paper offers a practical starting point, but additional research is necessary to establish the relevance of the chosen threshold by exploring the real-life meaning behind the data-driven threshold value. In particular, one should consider what life conditions are driving the data-cliff and whether these differ significantly across countries and population groups (e.g. males/females, old/young, urban/rural, etc.). It addition, one might consider adopting different cutoffs across countries – while this approach can be used for monitoring changes over time within countries, it is problematic for international comparisons of SWB.

The empirical portion of the paper expands on previous cross-country studies using regression analysis to explore the link between national SWB and objective indicators of development. It aims to contribute to the better understanding of this relationship in order to help inform future development policy. It recommends a Beta-regression approach over the baseline OLS model, and offers new insights into the measurement of SWB. The Beta-regression model improves the goodness-of-fit over the standard OLS models when using the share of satisfied individuals. An important advantage of using the non-linear Beta-regression model is that it can be used to assess non-constant relationships between different levels of SWB and objective measures, revealing differences along the progression paths of key measures of development. This econometric model does not establish causality; nevertheless, associations between SWB and objective measures of development are a useful tool for evaluating the welfare relevance of economic indicators. The insights gained from this exercise can help guide research into how economic wealth translates into overall well-being.

A principal finding is that the proportion of satisfied individuals is not significantly associated with *PER CAPITA GNI* when controlling for cultural differences, which contrasts the strong positive relationship between mean satisfaction and income. This finding does not invalidate the observed relationship between mean satisfaction and income, but it does reveal the importance of the aggregation approach used to measure national SWB and its implications for development policies. In light of this result, we should be skeptical about the benefits of raising income without considering distributional issues and clear social aims.

Lastly, a noteworthy question arises from the dichotomization of life satisfaction scales – should we instead focus on yes/no questions regarding whether the individual is sufficiently satisfied with life? It would be a worthwhile exercise to include such questions in future questionnaires alongside the standard life satisfaction scales in order to compare the two and gain further insight about a meaningful cut-off value. However, high resolution life satisfaction scales are important for individual-level analysis and the full distribution over these scales should be continually and rigorously monitored for any substantial changes – changes that may well affect the aggregation method altogether. The availability of high-resolution satisfaction data is vital for improving on current aggregation methods. However, binary questions can be useful validation tools for indirectly identifying meaningful cut-off points if they are included alongside high resolution scales. Alternatively, it may be feasible to formulate a question that directly identifies a subjective cut-off for each individual akin to the Minimum Income Question used to derive the Subjective Poverty Line (introduced by Goedhart et al., 1977), which asks individuals what minimum household income is needed to make ends meet.

Bibliography

- Abramovitz, M., 1959. The Welfare Interpretation of Secular Trends in National Income and Product. The Allocation of Economic Resources. Stanford University Press Stanford, California.
- Akerlof, G.A., Dickens, W.T., 1982. The Economic Consequences of Cognitive Dissonance. American Economic Review 72(3), 307-319.
- Bjornskov, C., Dreher, A., Fischer, J.A.V., 2008. Cross-Country Determinants of Life Satisfaction: Exploring Different Determinants Across Groups in Society. Social Choice Welfare 30, 119-173.
- Blanchflower, D.G., Bell, D.N.F., Montagnoli, A., Moro, M., 2014. The Happiness Trade-Off between Unemployment and Inflation. Journal of Money, Credit and Banking 46(2), 117-141.
- Blanchflower, D.G., Oswald, A.J., 2004. Well-being Over Time in Britain and the USA. Journal of Public Economics 88(7), 1359-1386.
- Blanchflower, D.G., Oswald, A.J., 2005. Happiness and the Human Development Index: The Paradox of Australia. Australian Economic Review 38(3), 307-318.
- Bond, T.N., Lang, K., 2014. The Sad Truth About Happiness Scales. National Bureau of Economic Research (NBER)
- Bruni, L., Comim, F., Pugno, M., 2008. Capabilities and Happiness. USA: Oxford University Press.
- Bruni, L., Porta, P.L., 2007. Handbook on the Economics of Happiness. USA: Oxford University Press.
- Burchardt, T., 2005. Are One Man's Rags Another Man's Riches? Identifying Adaptive Expectations Using Panel Data. Social Indicators Research 74, 57-102.
- Cameron, A.C., Trivedi, P.K., 2009. Microeconometrics Using Stata. Texas: Stata Press.
- Casacci, S., Pareto, A., 2018. Subjective Indicators Construction by Distance Indices: An Application to Life Satisfaction Data. Social Indicators Research 137(3), 831-846.
- Crisp, R., 2003. Equality, priority, and compassion. Ethics 113(4), 745-763.
- Cummins, R.A., 2016. Subjective Wellbeing as a Social Indicator. Social Indicators Research, 1-13.
- Cummins, R.A., Eckersley, R., Pallant, J., Van Vugt, J., Misajon, R., 2003. Developing a National Index of Subjective Wellbeing: The Australian Unity Wellbeing Index. Social Indicators Research 64, 159-190.
- Deaton, A., 2008. Income, health, and well-being around the world: Evidence from the Gallup World Poll. Journal of Economic Perspectives 22(2), 53-72.
- Di Tella, R., MacCulloch, R.J., Oswald, A.J., 2001. Preferences over inflation and unemployment: Evidence from surveys of happiness. American Economic Review 91(1), 335-341.
- Di Tella, R., MacCulloch, R.J., Oswald, A.J., 2003. The Macroeconomics of Happiness. Review of Economics and Statistics 85(4), 809-827.
- Diener, E., 1994. Assessing Subjective Well-Being: Progress and Opportunities. Social Indicators Research 31(2), 103-157.
- Diener, E., 2006. Guidelines for National Indicators of Subjective Well-Being and Ill-Being. Applied Research in Quality of Life 1(2), 151-157.
- Diener, E., Lucas, R.E., Schimmack, U., Helliwell, J.F., 2009. Well-being for Public Policy. New York: Oxford University Press.

- Diener, E., Oishi, S., 2000. Money and Happiness: Income and Subjective Well-being across Nations. In: Diener, E., Suh, E.M. (Eds.). Culture and Subjective Well-Being. MIT Press, pp. 185-218.
- Diener, E., Seligman, M.E.P., 2004. Beyond Money: Toward an Economy of Well-Being. Psychological Science in the Public Interest 5(1), 1-31.
- Diener, E., Suh, E., 1997. Measuring quality of life: Economic, social, and subjective indicators. Social Indicators Research 40(1-2), 189-216.
- Diener, E., Tov, W., 2012. National Accounts of Well-Being. In: Land, K.C., Michalos, A.C., Sirgy, M.J. (Eds.). Handbook of Social Indicators and Quality of Life Research. Springer New York, pp. 137-156.
- Easterlin, R.A., 1974. Does Economic Growth Improve the Human Lot? Some Empirical Evidence. In: David, A., Reder, M.W. (Eds.). Nations and Households in Economic Growth: Essays in Honour of Moses Abramovitz. Academic Press, Inc. New York, pp. 89-125.
- Easterlin, R.A., 2001. Income and Happiness: Towards a Unified Theory. The Economic Journal 111, 465-484.
- Easterlin, R.A., 2004. The Economics of Happiness. Daedalus 133(2), 26-33.
- Easterlin, R.A., 2013. Happiness and Economic Growth: The Evidence. Institue for the Study of Labor (IZA) Discussion Paper Series No. 7187.
- Easterlin, R.A., Angelescu McVey, L., Switek, M., Sawangfa, O., Zweig, J.S., 2011. The Happiness-Income Paradox Revisited. Institue for the Study of Labor (IZA) 5799.
- EU, 2014. Youth Unemployment in the EU: A Scarred Generation? , 12th Report of Session 2013-14. HL Paper 164. House of Lords European Union Committee.
- EVS, 2011. European Values Study 1981-2008, Longitudinal Data File ZA4804 Version 2.0.0 GESIS Data Archive: Cologne, Germany.
- Fattore, M., Maggino, F., Arcagni, A., 2015. Exploiting Ordinal Data for Subjective Well-Being Evaluation. Statistics in Transition New Series 16(3), 409-428.
- Ferrari, S., Cribari-Neto, F., 2004. Beta Regression for Modelling Rates and Proportions. Journal of Applied Statistics 31(7), 799-815.
- Ferrer-i-Carbonell, A., 2005. Income and Well-Being: An Empirical Analysis of the Comparison Income Effect. Journal of Public Economics 89, 997-1019.
- Fleurbaey, M., 2009. Beyond GDP: The Quest for a Measure of Social Welfare. Journal of Economic Literature 47(4), 1029-1075.
- Flouri, E., 2004. Subjective Well-Being in Midlife: The Role of Involvement of and Closeness to Parents of the Determinants of Happiness. Journal of Happiness Studies 5(4), 335-358.
- Frankfurt, H., 1987. Equality as a Moral Ideal. Ethics 98(1), 21-43.
- Frey, B.S., Stutzer, A., 2002a. Happiness and Economics. Princeton, NJ: Princeton University Press.
- Frey, B.S., Stutzer, A., 2002b. What Can Economists Learn from Happiness Research? Journal of Economic Literature 40(2), 402-435.
- Frey, B.S., Stutzer, A., 2014. Economic Consequences of Mispredicting Utility. Journal of Happiness Studies 15(4), 937-956.
- Goedhart, T., Halberstadt, V., Kapteyn, A., Van Praag, B., 1977. The Poverty Line: Concept and Measurement. Journal of Human Resources 12(4), 503-520.
- Graham, C., 2008. Some Insights on Development from the Economics of Happiness. In: Bruni, L., Comim, F., Pugno, M. (Eds.). Capabilities and Happiness. Oxford University Press New York, pp. 81-113.
- Graham, C., Pettinato, S., 2002. Happiness and Hardship: Opportunity and Insecurity in New Market Economies. Washington, DC: Brookings Institution Press.

- Hair, J.F., Black, W.C., Babin, B.J., Anderson, R.E., Tatham, R.L., 2006. Multivariate Data Analysis, 6th ed. New Jersey: Pearson Prentice Hall.
- Helliwell, J.F., 2003. How's Life? Combining Individual and National Variables to Explain Subjective Well-Being. Economic Modelling 20(2), 331-360.
- Helliwell, J.F., Barrington-Leigh, C.P., 2010. Viewpoint: Measuring and understanding subjective well-being. Canadian Journal of Economics-Revue Canadienne d Economique 43(3), 729-753.
- Helliwell, J.F., Huang, H., 2008. How's Your Government? International Evidence Linking Good Government and Well-Being. British Journal of Political Science 38(4), 595-619.
- Helliwell, J.F., Layard, R., Jeffrey, S., 2012. World Happiness Report. The Earth Institute: Columbia University.
- Hirschman, A.O., 1965. Obstacles to Development: A Classification and a Quasi-Vanishing Act. Economic Development and Cultural Change 13(4), 385-393.
- Inglehart, R., Welzel, C., 2010. Changing Mass Priorities: The Link Between Modernization and Democracy. Perspectives on Politics 9(2), 551-567.
- Kass, R.E., Raftery, A.E., 1995. Bayes Factors. Journal of the American Statistical Association 90(430), 773-795.
- Kesebir, P., Diener, E., 2008. In Defense of Happiness: Why Policymakers Should Care About Subjective Well-Being. In: Bruni, L., Comim, F., Pugno, M. (Eds.). Capabilities and Happiness. Oxford University Press New York, pp. 60-80.
- Kieschnick, R., McCullough, B.D., 2003. Regression Analysis of Variates Observed on (0, 1): Percentages, Proportions and Fractions. Statistical Modelling 3, 193-213.
- Kovacevic, M., 2010. Review of HDI Critiques and Potential Improvements. United Nations Development Programme
- Lawless, N.M., Lucas, R.E., 2011. Predictors of Regional Well-Being: A County Level Analysis. Social Indicators Research 101(3), 341-357.
- Layard, R., 2003. Happiness: Has Social Science a Clue?, Lionel Robbins Memorial Lectures 2002/03. London School of Economics.
- Layard, R., 2009. Why Subjective Well-Being Should Be the Measure of Progress, OECD World Forum: Charting Progress, Building Visions, Improving Life, Busan, Korea.
- Layard, R., 2011. Happiness : Lessons from a New Science. London: Penguin Books.
- Leigh, A., Wolfers, J., 2006. Happiness and the Human Development Index: Australia Is Not a Paradox. Australian Economic Review 39(2), 176-184.
- Maggino, F., 2009. Methodological Aspects and Technical Approaches in Measuring Subjective Well-Being.
- Mauro, V., Biggeri, M., Maggino, F., 2016. Measuring and Monitoring Poverty and Well-Being: A New Approach for the Synthesis of Multidimensionality. Social Indicators Research, 1-15.
- McGillivray, M., 2005. Measuring non-economic well-being achievement. Review of Income and Wealth (2), 337-364.
- Oreopoulos, P., 2003. Do Dropouts Drop Out Too Soon? International Evidence From Changes in School-Leaving Laws. National Bureau of Economic Research
- Oswald, A.J., 1997. Happiness and Economic Performance. The Economic Journal 107(445), 1815-1831.
- Ouweneel, P., Veenhoven, R., 1991. Cross-National Differences in Happiness: Cultural Bias or Societal Quality. In: Bleichrodt, N., Drenth, P.J.D. (Eds.). Contemporary Issues in Cross-Cultural Psychology. Swets and Zeitlinger Amsterdam.
- Ovaska, T., Takashima, R., 2006. Economic policy and the level of self-perceived well-being: An international comparison. Journal of Socio-Economics 35(2), 308-325.

- Powdthavee, N., 2008. Putting a Price Tag on Friends, Relatives, and Neighbors: using Surveys of Life-Satisfaction to Value Social Relationships. Journal of Socio-Economics 37(4), 1459-1480.
- Powdthavee, N., 2010. How Much Does Money Really Matter? Estimating the Causal Effects of Income on Happiness. Empirical Economics 39(1), 77-92.
- Powdthavee, N., Lekfuangfu, N., Wooden, M., 2015. What's the Good of Education On Our Overall Quality of Life? A Simultaneous Equation Model of Education and Life Satisfaction for Australia. Journal of Behavioral and Experimental Economics 54, 10-21.
- Shields, M.A., Wheatley-Price, S., Wooden, M., 2009. Life Satisfaction and the Economic and Social Characteristics of Neighborhoods. Journal of Population Economics 22(2), 421-443.
- Smithson, M., Verkuilen, J., 2006. A Better Lemon Squeezer? Maximum-Likelihood Regression with Beta-Distributed Dependent Variables. Psychological Methods 11(1), 54-71.
- Stevenson, B., Wolfers, J., 2008. Economic Growth and Subjective Well-Being: Reassessing the Easterlin Paradox. Brookings Papers on Economic Activity 2008, 1-87.
- Stiglitz, J.E., Sen, A., Fitoussi, J.-P., 2010. Report by the Commission on the Measurement of Economic Performance and Social Progress.
- UNDP, 2013. International Human Development Indicators Tables, United Nations Development Programme (UNDP).
- UNDP, 2016. Human Development for Everyone.
- Veenhoven, R., 1996. Happy life-expectancy. Social Indicators Research 39(1), 1-58.
- WDI, 2014. World Development Indicators Database, World Bank.
- WVS, 2009. World Values Survey 1981-2008 Official Aggregate v.20090901. World Values Survey Association., Aggregated File Producer: ASEP/JDS, Madrid, Spain.

Figures



Figure 1. Dissonance level across the SWB path



Figure 2. Distribution of life satisfaction responses, by wave



Figure 3. Cultural map (1999-2010 average).

Figure 4. Distribution characteristics of the proportion of satisfied individuals, by wave.



Note: measures computed using sampling weights.



Figure 5. Average marginal effects on mean satisfaction with 95% CIs (model 2)

Tables

	constructed waves			
Overall Life Satisfaction	1999	-2004	2005-2010	
1 - Dissatisfied	4,979	5.36%	4,809	3.34%
2	3,886	4.19%	3,286	2.28%
3	5,184	5.58%	6,511	4.52%
4	5,186	5.59%	7,465	5.19%
5	13,705	14.76%	17,521	12.17%
6	9,251	9.96%	14,637	10.17%
7	12,391	13.34%	22,134	15.38%
8	15,456	16.65%	30,545	21.22%
9	10,192	10.98%	16,943	11.77%
10 - Satisfied	11,576	12.47%	18,554	12.89%
missing	1,043	1.12%	1,515	1.05%

Table 1. Distribution of life satisfaction responses, by wave.

Source: WVS (2009), EVS (2011)

Note: Counts constructed using sampling weights.

Variable	Variable description	Source
Dependent variable		
SWB _{share}	Proportion of survey respondents who reported satisfaction level 5 or above.	WVS (2009), EVS (2011)
Objective development	indicators	
GNI PER CAPITA	Aggregate income of an economy generated by its production and its ownership of factors of production, less the incomes paid for the use of factors of production owned by the rest of the world, converted to (constant 2005) international dollars using purchasing power parity (PPP) rates, divided by midvear population.	UNDP (2011)
LIFE EXPECTANCY	Number of years a newborn infant could expect to live if prevailing patterns of age-specific mortality rates at the time of birth stay the same throughout the infant's life.	UNDP (2011)
MEAN YEARS OF SCHOOLING	Average number of years of education received by people ages 25 and older, converted from education attainment levels using official durations of each level.	UNDP (2011)
EXPECTED YEARS OF SCHOOLING	Number of years of schooling that a child of school entrance age can expect to receive if prevailing patterns of age-specific enrolment rates persist throughout the child's life	UNDP (2011)
Cultural controls		
Index of traditional/ secular-rational	Constructed using several WVS and EVS attitudinal questions and measured as a continuous scale, where countries dominated by more traditional values are given negative scores, and countries with more secular-rational values are given a positive score. Larger negative (positive) scores reflect stronger traditional (secular-rational) values. Traditional values emphasize religion, national pride, obedience and respect for authority. Secular-rational values emphasize secularism, cosmopolitanism, autonomy, and rationality.	Inglehart and Welzel (2010)
Index of survival/ self-expression	Constructed using several WVS and EVS attitudinal questions and measured as a continuous scale, where countries dominated by survival values are given negative scores, and countries that value self-expression more are given a positive score. Larger negative (positive) scores reflect stronger survival (self-expression) values. Survival values emphasize order, economic security and conformity. Self-expression values emphasize the importance of self-expression, participation, subjective well-being, trust, tolerance, and quality of life.	Inglehart and Welzel (2010)
Macro-level controls Unemployment	Share of the labour force that is without work but available for and seeking employment (national estimate).	WDI (2014)
Inflation	Annual percentage change in the cost to the average consumer of acquiring a basket of goods and services, measured by the consumer price index.	WDI (2014)

Table 2. Variable definitions.

	mean	st. dev.	min.	max.	observations
SWB _{share} (0-1)					
1999-2004	0.80	0.15	0.39	0.98	63
2005-2010	0.85	0.11	0.54	0.98	78
total	0.83	0.13	0.39	0.98	141
GNI PER CAPITA (PPP cons	stant 2005 \$)				
1999-2004	15,100	12,417	608	53,204	63
2005-2010	17,548	13,244	809	53,763	78
total	16,454	12,894	608	53,763	141
LIFE EXPECTANCY (years)					
1999-2004	71.89	7.68	44.70	81.20	63
2005-2010	73.81	7.70	46.94	82.86	78
total	72.95	7.72	44.70	82.86	141
MEAN YEARS OF SCHOOLIN	NG				
1999-2004	8.38	2.43	3.30	13.00	63
2005-2010	9.03	2.69	1.30	12.66	78
total	8.74	2.59	1.30	13.00	141
EXPECTED YEARS OF SCHO	DOLING				
1999-2004	13.20	2.78	5.40	18.00	63
2005-2010	13.82	2.64	5.64	18.00	78
total	13.54	2.72	5.40	18.00	141
Unemployment					
1999-2004	9.16	5.44	2.32	26.08	63
2005-2010	8.62	5.68	1.23	34.37	75
total	8.86	5.56	1.23	34.37	138
Inflation					
1999-2004	8.93	23.22	-0.49	174.21	61
2005-2010	5.23	4.02	-0.11	17.25	77
total	6.87	15.76	-0.49	174.21	138

Table 3. Summary statistics.

Note: Life satisfaction statistics computed using raw data with no sampling weights applied.

	(1)	(2)	(3)
ln(GNI PER CAPITA)	0.0568 ***	0.0154	0.0124
	(0.0146)	(0.0112)	(0.0110)
LIFE EXPECTANCY	0.0029 *	0.0034 ***	0.0026 **
	(0.0016)	(0.0011)	(0.0012)
MEAN YEARS OF SCHOOLING	-0.0218 ***	-0.0085 *	-0.0079
	(0.0050)	(0.0048)	(0.0052)
EXPECTED YEARS OF	0.0132 ***	0.0076	0.0102 *
SCHOOLING	(0.0044)	(0.0050)	(0.0052)
wave dummy	0.0266 **	0.0313 ***	0.0312 ***
	(0.0107)	(0.0091)	(0.0093)
index of traditional/secular-		-0.0077	-0.0102
rational values		(0.0086)	(0.0089)
index of survival/self-		0.0617 ***	0.0573 ***
expression values		(0.0075)	(0.0079)
unemployment			-0.0016 *
			(0.0008)
inflation			-0.0005 **
			(0.0002)
BIC	-321	-384	-363
Observations	141	141	136

 Table 4. Beta-regression marginal effects at means for share of satisfied individuals.

 dependent variable: share of satisfied individuals (SWBsterm)

*** p<0.01, ** p<0.05, * p<0.1, panel-robust standard errors in parentheses.

All regressions include a constant term (not shown here). Share of satisfied individuals constructed using sampling weights.

Table 5.

dependent variable:	share of satisf	ied individuals	share of satisfied individuals		share of satisfied	share of satisfied
	(1)	(2)	(3)	(4)	women	men
	wave 1	wave 2	developed	developing	(5)	(6)
ln(GNI PER CAPITA)	-0.0087	0.0327 **	0.0748 ***	0.0147	0.0200 *	0.0111
	(0.0155)	(0.0129)	(0.0193)	(0.0157)	(0.0110)	(0.0120)
LIFE EXPECTANCY	0.0068 ***	0.0009	0.0009	0.0044 ***	0.0029 ***	0.0039 ***
	(0.0017)	(0.0011)	(0.0028)	(0.0013)	(0.0011)	(0.0011)
MEAN YEARS OF SCHOOLING	-0.0135 *	-0.0052	0.0066	-0.0137 *	-0.0087 *	-0.0086 *
	(0.0076)	(0.0042)	(0.0045)	(0.0082)	(0.0050)	(0.0048)
EXPECTED YEARS OF SCHOOLING	0.0129	0.0045	0.0085 **	0.0112	0.0073	0.0080 *
	(0.0087)	(0.0043)	(0.0034)	(0.0098)	(0.0053)	(0.0048)
index of traditional/secular-rational values	-0.0065	-0.0108	-0.0211 **	0.0146	-0.0101	-0.0054
	(0.0136)	(0.0073)	(0.0083)	(0.0161)	(0.0084)	(0.0089)
index of survival/self-expression values	0.0787 ***	0.0504 ***	0.0198 **	0.1113 ***	0.0603 ***	0.0624 ***
	(0.0139)	(0.0063)	(0.0081)	(0.0191)	(0.0073)	(0.0080)
wave dummy	ye	es	yes		yes	yes
development dummy	n	0	yes		no	no
interaction	wave intera	acted w/ all	developed interacted w/ all		none	none
DIG	covar	riates	covariates		201	
BIC	-3	65	-3	68	-381	-372
Observations	141		1	141		141

*** p<0.01, ** p<0.05, * p<0.1, panel-robust standard errors in parentheses. All regressions include a constant term (not shown here). Share of satisfied individuals constructed using sampling weights.

	difference in share of satisfied individuals	difference in mean satisfaction
Bulgaria	0.073 (0.020) ***	0.611 (0.104) ***
Cyprus	-0.017 (0.013)	-0.008 (0.097)
Finland	-0.008 (0.011)	-0.115 (0.080)
France	-0.011 (0.013)	0.172 (0.082) **
Georgia	0.098 (0.016) ***	0.528 (0.088) ***
Germany	-0.026 (0.011) **	-0.028 (0.069)
Great Britain	-0.060 (0.010) ***	-0.101 (0.074)
Italy	-0.034 (0.012) ***	0.256 (0.080) ***
Moldova	0.135 (0.018) ***	1.138 (0.097) ***
Netherlands	0.003 (0.005)	0.257 (0.054) ***
Norway	-0.015 (0.008) *	0.149 (0.074) **
Poland	-0.008 (0.013)	0.187 (0.087) **
Romania	0.105 (0.015) ***	1.028 (0.090) ***
Russia	0.036 (0.015) **	0.429 (0.088) ***
Slovenia	-0.019 (0.010) *	0.301 (0.083) ***
Spain	-0.036 (0.009) ***	-0.005 (0.064)
Sweden	-0.055 (0.012) ***	-0.112 (0.084)
Switzerland	-0.031 (0.009) ***	0.002 (0.071)
Turkey	-0.137 (0.013) ***	-0.958 (0.087) ***
Ukraine	0.070 (0.021) ***	0.410 (0.111) ***

 Table 6. T-tests for differences in aggregate SWB between EVS and WVS samples for countries

 surveyed under both initiatives in wave 2⁺/₁

*** p<0.01, ** p<0.05, *p<0.1; standard errors in parantheses.
† t-test conducted using sample weights (a positive point estimate indicates an increase</pre> in aggregate SWB from 2005-2007 (the WVS sample) to 2008-2010 (the EVS sample).

Appendix

	wave 1	wave 2		wave 1	wave 2
	(1999-2004)	(2005-2010)		(1999-2004)	(2005-
Albania	\checkmark	\checkmark	Kyrgyzstan	\checkmark	
Algeria	\checkmark		Latvia	\checkmark	\checkmark
Andorra		\checkmark	Lithuania	\checkmark	\checkmark
Argentina	\checkmark	\checkmark	Luxembourg	\checkmark	\checkmark
Armenia		\checkmark	Macedonia	\checkmark	\checkmark
Australia	\checkmark	\checkmark	Malaysia		\checkmark
Austria	\checkmark	\checkmark	Mali		\checkmark
Azerbaijan		\checkmark	Malta	\checkmark	\checkmark
Bangladesh	\checkmark		Mexico	\checkmark	\checkmark
Belarus	\checkmark	\checkmark	Moldova	\checkmark	\checkmark
Belgium	\checkmark	\checkmark	Morocco	\checkmark	\checkmark
Bosnia-Herzegovina	\checkmark	\checkmark	Netherlands	\checkmark	\checkmark
Brazil		\checkmark	New Zealand		\checkmark
Bulgaria	\checkmark	\checkmark	Nigeria	\checkmark	
Burkina Faso		\checkmark	Norway		\checkmark
Canada	\checkmark	\checkmark	Pakistan	\checkmark	
Chile	\checkmark	\checkmark	Peru	\checkmark	\checkmark
China	\checkmark	\checkmark	Philippines	\checkmark	
Colombia		\checkmark	Poland	\checkmark	\checkmark
Croatia	\checkmark	\checkmark	Portugal	\checkmark	\checkmark
Cyprus		\checkmark	Republic of Korea	\checkmark	\checkmark
Czech Republic	\checkmark	\checkmark	Romania	\checkmark	\checkmark
Denmark	\checkmark	\checkmark	Russian Federation	\checkmark	\checkmark
Egynt	1	1	Rwanda		1
Estonia	✓ ✓	✓ ✓	Saudi Arabia	\checkmark	
Ethionia		1	Serbia & Montenegro	\checkmark	\checkmark
Finland	\checkmark	✓ ✓	Singanore	✓ ✓	
France	✓ ✓	✓ ✓	Slovakia	✓ ✓	\checkmark
Georgia		1	Slovenia	\checkmark	1
Germany	\checkmark	✓ ✓	South Africa	✓ ✓	✓ ✓
Ghana		1	South I finite	\checkmark	1
Great Britain & N Ireland	\checkmark	✓ ✓	Sweden	✓ ✓	✓ ✓
Greece	✓ ✓	✓ ✓	Switzerland	,	✓ ✓
Guatemala	ŗ	√ √	Tanzania	\checkmark	•
Hong Kong		, ,	Thailand	•	\checkmark
Hungary	1	√ √	Trinidad & Tobago		, ,
Iceland	1	1	Turkey	1	, ,
India	1	v	Landa	· ·	•
Indonesia	· ·	· ·	Ultraine	· ·	1
Iran	· ·	· ·		·	·
Iran	•	·	U.S.	•	·
II ay Irolond	1	* ./	Veneruala	1	•
Icroal	•	¥	v chezuela Viat Nam	•	1
Israel	*		viet inam Zambia	v	•
Italy	*	•	Zambakara		v
Japan	v	v	Zimbaowe	v	
Jordan	v	v			

Table A1. Country availability (by wave).

		Mean	St. Dev.
transformed mean satisfaction (ranges 0-1)	overall	0.633	0.114
	between		0.114
	within		0.033
share of satisfied individuals (ranges 0-1)	overall	0.828	0.128
	between		0.129
	within		0.039
<i>ln(</i> GNI PER CAPITA)	overall	9.263	1.098
	between		1.183
	within		0.129
LIFE EXPECTANCY	overall	72.954	7.720
	between		8.612
	within		0.918
MEAN YEARS OF SCHOOLING	overall	8.744	2.593
	between		2.754
	within		0.391
EXPECTED YEARS OF SCHOOLING	overall	13.542	2.716
	between		2.879
	within		0.500
total number of observations $= 141$			

Table A2. Decomposed variance statistics for measures of interest

dependent variable: share of satisfied individuals (SWB _{share})						
	(1)	(2)	(3)			
ln(GNI PER CAPITA)	0.0688 ***	0.0323 *	0.0271			
	(0.0192)	(0.0167)	(0.0165)			
LIFE EXPECTANCY	0.0041 *	0.0045 **	0.0029			
	(0.0024)	(0.0020)	(0.0020)			
MEAN YEARS OF SCHOOLING	-0.0247 ***	-0.0129 **	-0.0119 *			
	(0.0060)	(0.0057)	(0.0061)			
EXPECTED YEARS OF SCHOOLING	0.0132 **	0.0075	0.0114 *			
	(0.0052)	(0.0054)	(0.0059)			
wave dummy	0.0362 ***	0.0368 ***	0.0352 ***			
	(0.0121)	(0.0112)	(0.0116)			
index of traditional/secular-		-0.0122	-0.0141			
rational values		(0.0095)	(0.0096)			
index of survival/self-expression		0.0522 ***	0.0468 ***			
values		(0.0089)	(0.0092)			
unemployment			-0.0016			
			(0.0013)			
inflation			-0.0013 ***			
			(0.0004)			
BIC	-274	-306	-291			
Observations	141	141	136			

Table A3. OLS coefficients for share of satisfied individuals.

*** p<0.01, ** p<0.05, * p<0.1, panel-robust standard errors in parentheses.

All regressions include a constant term (not shown here). Share of satisfied individuals constructed using sampling weights.

dependent variable: mean satisfac	lependent variable: mean satisfaction (SWBmean)						
	(1)	(2)	(3)				
ln(GNI PER CAPITA)	0.0613 ***	0.0215 *	0.0185 *				
	(0.0146)	(0.0111)	(0.0109)				
LIFE EXPECTANCY	0.0026	0.0032 **	0.0023 *				
	(0.0017)	(0.0013)	(0.0013)				
MEAN YEARS OF SCHOOLING	-0.0195 ***	-0.0041	-0.0036				
	(0.0046)	(0.0046)	(0.0049)				
EXPECTED YEARS OF	0.0126 ***	0.0067	0.0083 *				
SCHOOLING	(0.0043)	(0.0046)	(0.0047)				
wave dummy	0.0251 **	0.0267 ***	0.0284 ***				
	(0.0105)	(0.0089)	(0.0091)				
index of traditional/secular-		-0.0218 **	-0.0216 **				
rational values		(0.0091)	(0.0093)				
index of survival/self-		0.0568 ***	0.0555 ***				
expression values		(0.0069)	(0.0072)				
unemployment			-0.0007				
			(0.0010)				
inflation			-0.0007 ***				
			(0.0002)				
BIC	-305	-367	-348				
Observations	141	141	136				

Tał	h	A4	Ret	a-regres	sion n	narginal	effects at	means	for	mean	satisfa	ction
1 a	лс	A4.	Deta	a-i egi es	51011 II	naigmai	cifects at	incans.	101	шсап	Sausia	CUOH

*** p<0.01, ** p<0.05, * p<0.1, panel-robust standard errors in parentheses.

All regressions include a constant term (not shown here). Share of satisfied individuals constructed using sampling weights.

dependent variable: share of satisfied individuals (SWB _{share})						
	wave 1	wave 2				
ln(GNI PER CAPITA)	0.0053	-0.0040				
	(0.018)	(0.017)				
LIFE EXPECTANCY	0.0058 ***	0.0012 *				
	(0.001)	(0.001)				
MEAN YEARS OF SCHOOLING	-0.0091	-0.0044				
	(0.007)	(0.005)				
EXPECTED YEARS OF SCHOOLING	0.0052	0.0110 **				
	(0.007)	(0.005)				
index of traditional/secular-rational values	-0.0059	-0.0037				
	(0.013)	(0.008)				
index of survival/self-expression values	0.0786 ***	0.0522 ***				
	(0.012)	(0.008)				
wave dummy	yes					
development dummy	no					
interaction	wave interacted w/ all covariates					
BIC	-264					
Observations		102				

Table A5. Beta-regression marginal effects at means by wave for subsample of countries that appear in both waves.

*** p<0.01, ** p<0.05, * p<0.1, panel-robust standard errors in parentheses. All regressions include a constant term (not shown here). Share of satisfied individuals constructed using sampling weights.